Author’s response to reviews

Title: MULTI-CRITERIA ANALYSIS IN THE HEALTH AREA: SELECTION OF THE MOST APPROPRIATE TRIAGE SYSTEM FOR THE EMERGENCY CARE UNITS IN NATAL.

Authors:
Deyse Camilo (deysecamilo@gmail.com)
Ricardo Souza (RICARDOPIRES@CT.UFRN.BR)
Talita Frazão (thalytachaggas@gmail.com)
João Junior (jfcj1977@gmail.com)

Version: 2 Date: 06 Feb 2020

Author’s response to reviews:

UNIVERSIDADE FEDERAL DO RIO GRANDE DO NORTE – UFRN

Centro de Tecnologia, Departamento de Engenharia de Produção
Avenida Senador Salgado Filho - de 2235 a 4501 - lado ímpar,
Capim Macio, 59080000 - Natal, RN - Brasil

Natal, February 03, 2020

A. Cuff
Editor
BMC Medical Informatics and Decision Making

Ref.: Paper: “MULTI-CRITERIA ANALYSIS IN THE HEALTH AREA: SELECTION OF THE MOST APPROPRIATE TRIAGE SYSTEM FOR THE EMERGENCY CARE UNITS IN NATAL”.


Dear Gilherme Del Fiol,

Thank you very much for your careful review and constructive suggestions regarding our manuscript. Those comments were quite valuable for us to revise it and improve its quality.

As a result of that, we would like to re-submit the manuscript to the BMC Medical Informatics and Decision Making.
A detailed response to all comments made by the Referee can be found below. Have you any further queries; please do not hesitate to contact us directly.

We look forward to hearing from you soon.

Yours Sincerely

D. G. G. Camilo

Correspondent author

---

Guilherme Del Fiol (Editor Comments)

Comment: The editor suggested a change in the title, replacing the word “MULTICRITERARY” with “MULTI-CRITERIA”.

Answer: We answered the publisher's request and the title went like this: "MULTI-CRITERIA ANALYSIS IN THE HEALTH AREA: SELECTION OF THE MOST APPROPRIATE TRIAGE SYSTEM FOR THE EMERGENCY CARE UNITS IN NATAL."